Author's response to reviews

Title: Meal frequencies in early adolescence predict meal frequencies in late adolescence and early adulthood

Authors:

Trine P Pedersen (Tppe@niph.dk)
Bjørn E Holstein (bho@niph.dk)
Mette Rasmussen (mera@niph.dk)

Version: 2 Date: 29 June 2012

Author's response to reviews:

Cover letter

Dear Editor,

We hereby send you our manuscript: ”Meal frequencies in early adolescence predict meal frequencies in late adolescence and early adulthood”. We hope you will consider this manuscript for publication in your journal. The topic and result of the study is relevant for the journal's audience and highly relevant in relation to the aims and scope of your journal. Furthermore, our manuscript covers an under-investigated topic, and the findings are unique.

Kind regards

Trine Pagh Pedersen
PhD Student

Competing interests

The authors, Trine Pagh Pedersen, Bjørn Holstein and Mette Rasmussen declare that there are none financial competing interest or non-financial competing interest.